Author's response to reviews

Title: Texture analysis on MR images helps predicting non-response to NAC in breast cancer

Authors:

Nicolas Michoux (nicolas.michoux@uclouvain.be)
Stéphane van den Broeck (stephane.vandenbroeck@gmail.com)
Laure Lacoste (lacostelaure@hotmail.com)
Latifa Fellah (latifa.fellah@uclouvain.be)
Christine Galant (christine.galant@uclouvain.be)
Martine Berlière (martine.berliere@uclouvain.be)
Isabelle Leconte (isabelle.leconte@uclouvain.be)

Version: 10 Date: 13 July 2015

Author's response to reviews: see over
Dear Dr Paolo Bruzzi,

It is my pleasure to hereby submit the revised version of our manuscript entitled "Texture analysis of MR images helps predicting non-response to NAC in breast cancer" for publication in BMC Cancer.

We provided a positive response to reviewer 4 and modified the text accordingly.

For all subsequent correspondence, please write me at the above address.

Yours Sincerely,

Nicolas Michoux, PhD.

Conflict of interest: none in respect to this work and its publication.
Financial interest: none in respect to this work and its publication.
All authors have agreed to the submission to the journal and that the manuscript is not currently under submission in any other journal.
Detailed response

Reviewer 4

1. The author responded satisfactorily to the comments. He made the corrections that were feasible to perform. I recommend to modify in the statistical analysis, that numerical variables will be describe through median and range, so there is no inconsistency with the results.

As suggested, data are now described in terms of median and range (95%CI on the median) (see first paragraph of the Statistical analysis).